Author's response to reviews

Title: Design of the Physical exercise during Adjuvant Chemotherapy Effectiveness Study (PACES): A randomized controlled trial to evaluate effectiveness and cost-effectiveness of physical exercise in improving physical fitness and reducing fatigue

Authors:

Hanna van Waart (h.v.waart@nki.nl)
Martijn M Stuiver (m.stuiver@nki.nl)
Wim H van Harten (w.v.harten@nki.nl)
Gabe S Sonke (g.sonke@nki.nl)
Neil K Aaronson (n.aaronson@nki.nl)

Version: 5 Date: 6 December 2010

Author's response to reviews:

One of the references has been withdrawn, since that manuscript has not yet been published.